Mini MOSS Debrief Minutes

June 16, 2004

ELEMENT ATTENDANCE:

ESDIS SSDO and System Manager, FDS (provided slides), EDOS, EMOS, ASDC DAAC, TES SIPS, GES DAAC, MLS SIPS, HIRDLS SIPS, OMI SIPS, MRTT

Mini MOSS 2: June 29, 2004

NEXT TELECON:  As-needed basis

DIAL-IN NUMBER:

USA Toll Free Number: 866-581-4760

USA Toll Number: +1-210-280-9695 (international)

PASSCODE: 574686
MEETING HIGHLIGHTS:

All debrief presentations can be obtained from the Aura Testing website: http://jupiter02.gsfc.nasa.gov/AuraTesting/ under Mini-MOSS link. Only highlights, changes, or clarifications will be presented below.

SSDO:

Karen reviewed the goals set for the test and mentioned that we fail short of some of them. We were able to test Release 7 at the DAACs and at launch configurations at the SIPS but some problems were encountered. All APIDs were re-tested and Averstar tested Hidden EDG. Scott re-ordered all products for HIRDLS and MLS. We were not able to re-test as products as we had hoped and there were a few failures. Secure MTMGW was not exercised. Due to many elements experiencing problems during the Mini MOSS, Mini MOSS 2 was proposed (initially requested by ODPS/OMI SIPS). June 28, 29 or 30th dates were discussed. After the telecon, Karen sent out an email stating that June 29th is the planned date for Mini MOSS 2.

EDOS:

Ron reported that EDOS experienced a few problems, none related to EDOS operations. They had to change their password for OMI NRTS access. During normal operations, the OA procedures would have to be followed in order to have the change made ahead of time. After four SCSs, EDOS restarted the test due to processing conflicts created by another test, which ran the previous day. Once re-started, no issues were encountered. EDOS re-started the test around 10:30am EST. This resulted in the elements receiving some duplicates. Ron also mentioned that he put in a request today to get HIRDLS APID 1616 for EDS installed into the system for use for Mini MOSS 2. It was also confirmed that for Mini MOSS 2, the transfer of the SCSs would be ~2 SCSs per hour, same as Mini MOSS (1).

EMOS:

Paul presented the slides via telecon. Paul reported on his Issues and Problems Found slide that it was observed that the Attitude Carryout files took longer to create than expected. After further discussions, it was determined that carryout files arrived at the GES DAAC according to the normal operational timeline, no delays. He also mentioned that he spoke with Mike Anderson regarding the orbit event files and Mike manually delivered the files (to OMI SIPS) late Monday (14th) and Elaine confirmed during the telecon that they had received the files. 

ASDC DAAC: 

Sue presented the slide via telecon. She mentioned in her overall status that ASDC OPS Release 7 (Synergy III mode) successfully ingested TES L0/DPREP data and distributed the data via secure copy to TES SIPS at 2-3x the operational data rate with MISR doing 5x processing. Sue mentioned that the Secure Copy Distribution improvement table column headings were incorrect, interchanged. The slides were corrected and re-posted after the telecon. ASDC received all the expected data and two TES L2 products failed ingest due to metadata validation error (wrong short names).

TES SIPS: 
Andy presented the slides via telecon. He reported that the duplicate L0 and attitude data caused initial delays as well as processing failures over the weekend. PGE failure occurred Friday evening and it was not caught until Monday morning. Andy reported GOOD NEWS for 4x transfer rate during the test. TES data totals included the duplicates received. He also pointed out an error in his Higher Level Products chart (slide 4), TL2ATMTN and TL2ATMTL products failed because of a bad short name. He is working with the developers on this. Should be a quick fix. 

TES would like to deliver new test data for a subset of APIDs for use in the Mini MOSS 2. It’s ~45GB of data. They will keep the MOSS 4 version for the remaining APIDs.

GES DAAC:

Ed presented the GES DAAC slide via telecon. He reported on his issues slide that the DAAC experienced a severe backlog of queues leading into the Spatial Subscription Server (~8000 in queue). They have since received an engineering fix (NCR 40669) to solve the problem. During the Mini MOSS, the DAAC bounced the server and received temporary relief. Ed confirmed with MLS SIPS that they were missing a granule for a few APIDs (ENG5 & 6, and SCI 1 &2). Vince from HIRDLS informed ED that they also delivered QA, Browse and PH files with their L2 product. Ed will look for the files in the archive. GES DAAC also received L1s and L3s (Daily) from MLS and numerous L1s from OMI SIPS. 

MLS:
Audrey presented the slides via telecon. She reported that due to hardware problems encountered during maintenance of the operational system, processing of data and staging of products were delayed. All problems were resolved. MLS performed L1 processing at the SIPS and at the SCF using the relayed data from the SIPS and L3 daily processing at the SIPS using simulated inputs. Products were ingested at the GES DAAC. MLS data totals included the duplicates received. Audrey mentioned that they did not get a chance to test thoroughly the Secure MTMGW; need to test on their side. Karen asked Audrey if she could provide more information on the hardware problems encountered and the delay. Audrey mentioned that the problem had to do with a video card being replaced as well as checking out the SIPS after their system upgrade. Audrey mentioned that they started processing the L3s on June 11th and the L1s were re-planned afterwards.

OMI SIPS:

Elaine presented the slides for OMI via telecon. She reported that OMI encountered the problem that the orbit number was not being converted correctly; a fix was applied and resolved the problem. They re-run Mission Test #2 data to test the fix. They also experienced L1B integration problems. OMI received all their expected files. Elaine updated their slides to reflect the receipt of the EDSs and orbit event files and the slides were re-posted to the web site.

HIRDLS:

Vince presented the slides via telecon.  Vince mentioned that the conversion to anonymous FTP push was successful. However, tighter security controls on anonymous FTP mean that Goddard DAAC cannot see, overwrite or delete files, making it harder for them to troubleshoot problems. This is admittedly inconvenient for all, but necessary.  (FTP pull looks better all the time.) Ed stated that he would talk to Bruce Vollmer about HIRDLS recommendation.
Vince was awaiting confirmation of RBD deliveries to Oxford. (UCB didn’t ask for Oxford’s results until yesterday--6/15/04). HIRDLS received all their expected files plus duplicates and produced all of their higher-level products.

HIRDLS is going to look at trying to re-deliver new science test data for Mini MOSS 2. Vince is to let Karen know in about two days.  APID 1616 EDS is planned for Mini MOSS 2. GES DAAC has the ESDT in TS2 and will promote to OPS in a few days. 
FDS: Slides posted on the web site.

ODPS: Slides posted on the web site. 
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